Author's response to reviews


Authors:

Lieke A.M.W. Wijnhoven (l.wijnhoven@pwo.ru.nl)
Daan H.M. Creemers (d.creemers@ggzoostbrabant.nl)
Rutger C.M.E. Engels (rengels@trimbos.nl)
Isabela Granic (i.granic@pwo.ru.nl)

Version: 2
Date: 5 June 2015

Author's response to reviews: see over
Dear Sir / Madam,

Please find enclosed the revised manuscript that I would like to submit for publication in BMC Psychiatry, entitled “The effect of the video game Mindlight on anxiety symptoms in children with an Autism Spectrum Disorder”. Co-authors are dr. Daan Creemers, Prof. dr. Rutger Engels and Prof. dr. Isabela Granic. The manuscript contains a total of 23 pages.

The following changes are made to the original manuscript:
- Trial registration date (line number 57):
  The trial registration date has been added to the trial registry and the unique identifying number.
- Consent (line number 194 and 196):
  It has been clarified that the active informed consent of parents and children above the age of 12 is written.
- Authors Contributions (line number 461-466):
  The details of the contributions of all authors to the final manuscript are clarified in the Authors Contribution.
- Funding (line number 468-472):
  In the Acknowledgements, it has been confirmed that the funding body has conducted peer review of the study protocol and that they have given approval to the study.

The ethical approval letter and the funding approval letter are sent as email attachments to the email address BMCSeriesEditorial@biomedcentral.com. We hope that the revision of the manuscripts and the additional information meet the journal’s requirements.

We look forward to receiving your response.

Yours sincerely,

Lieke Wijnhoven

Behavioural Science Institute
Radboud University Nijmegen
P.O. Box 9104
6500 HE Nijmegen
The Netherlands
Telephone: 00 31 24 361 16 14
E-mail: l.wijnhoven@pwo.ru.nl